Author's response to reviews

Title: Injection practice in Kaski district, Western Nepal: A community perspective

Authors:

Sudesh Gyawali (sudeshgy@hotmail.com)
Devendra S Rathore (rathoredsp66@gmail.com)
Pathiyil R Shankar (ravi.dr.shankar@gmail.com)
Vikash K KC (ykkc2001@gmail.com)
Manisha Maskey (manzasharma@gmail.com)
Nisha Jha (nishajha32@gmail.com)

Version: 2  Date: 17 July 2014

Author's response to reviews: see over
To
The Editor
BMC Public Health

**Subject: Submission of Manuscript for publication**

Dear Editor,

We are submitting a manuscript entitled “Injection practice in Kaski district, Western Nepal: A community perspective” for consideration of publication in your esteemed journal as an original article. The study quantifies injection usage, identifies the personnel who provide injections, and explores peoples’ knowledge and perception about injection use among the people of Kaski district. The data were collected and compared between urban and rural areas of Kaski district, western Nepal. The injection practice study at the community level is the first study of this type in Nepal. The findings of the study might be valuable in planning evidence-based interventions to make the injection practice safer in Nepal and might be of interest to the readers from both developed and developing countries. On behalf of all the contributors, I will act as guarantor and will correspond with the journal from this point onward.

The manuscript has not been submitted nor published or accepted for publication in any other journal. We have no competing interests.

As we are not capable of paying the publication charge (belong to low income developing country), we humbly request you to waive the publication charges if the manuscript is accepted.

Thanking you,

Yours’ sincerely,

Sudesh Gyawali & coauthors
Assistant Professor, Department of Pharmacology
Manipal College of Medical Sciences (MCOMS)
P.O. Box: 155, Pokhara, Nepal
Phone: 00977-61-440260, 00977-9848032051 (Mob.)
E-mail: sudeshgy@hotmail.com